Author's response to reviews

Title: A prevalence survey of every-day activities in pregnancy

Authors:

Samantha J Lain (samantha.lain@sydney.edu.au)
Jane B Ford (jane.ford@sydney.edu.au)
Ruth M Hadfield (rhadfield@med.usyd.edu.au)
Christine L Roberts (christine.roberts@sydney.edu.au)

Version: 3 Date: 13 July 2010

Author's response to reviews: see over
13th July 2010

Prof Cecily Begley  
The BioMed Central Editorial Team  
BMC Pregnancy and Childbirth  

Dear Professor Begley,

Attached please find the revised manuscript entitled “A prevalence survey of everyday activities in pregnancy”. Thank you for accepting it, in principle, for publication in BMC Pregnancy and Childbirth. The main text contains 1,899 words.

We have made the changes on pages 4 and 9 as suggested by the Associate Editor. We have also checked the manuscript and believe that it conforms to the manuscript checklist on the BioMed Central website.

If you have any queries regarding the manuscript please do not hesitate to contact me.

Yours sincerely

Samantha Lain & Dr Ruth Hadfield  
Address for Correspondence:  
Samantha Lain  
Level 2, Building 52.  
c/- Dept of O&G  
Royal North Shore Hospital, St Leonards 2065  
email: samlain@med.usyd.edu.au